Author's response to reviews

Title: Symptom attributions in patients with colorectal cancer: a population based study

Authors:

Line F Jensen (line.jensen@ph.au.dk)
Line Hvidberg (line.hvidberg@ph.au.dk)
Anette F Pedersen (afp@feap.dk)
Peter Vedsted (p.vedsted@feap.dk)

Version: 4 Date: 4 July 2015

Author's response to reviews: see over
Dear Editor,


Thank you for the opportunity to revise this manuscript and thank you for some valuable comments made by each of the reviewer. We have now revised the manuscript and this cover letter provides a detailed description of our responses to each of the points made by the reviewers.

We hope that this will sufficiently answer the raised issues and the article will be suitable for publication. Do not hesitate to contact us if there are any further issues to be handled.

On behalf of the authors,

Yours sincerely,

Line Flytkjær Jensen
PhD Fellow
RE. Reviewer 1:

1) Discretionary Revisions: Table 1 - a column with the total number of subjects per row could be added

This has now been added to table 1.

2) Abstract - The first sentence in the conclusions is not very clear and should be reworded to communicate the message more clearly

This has now been changed and now reads:

3) Needs some language corrections before being published

The manuscript has now undergone formal language revision.

RE. Reviewer 2:

No comments were made by reviewer 2.